Author’s response to reviews

Title: Damage to Sebaceous Gland and the Efficacy of Moisturizer after Whole Breast Radiotherapy: a Randomized Controlled Trial

Authors:

Mami Ogita (o-tamami@hotmail.co.jp)
Kenji Sekiguchi (kenjisek@luke.ac.jp)
Keiko Akahane (keiko-a@omiya.jichi.ac.jp)
Ryoko Ito (ryokoiwa@luke.ac.jp)
Chiori Haga (chiohaga@luke.ac.jp)
Satoru Arai (arasato@luke.ac.jp)
Yasushi Ishida (yaishida2009@yahoo.co.jp)
Jiro Kawamori (jirokawa@luke.ac.jp)

Version: 1 Date: 14 Jan 2019

Author’s response to reviews:

Dear editor and reviewers,

We would like to thank the editor and reviewers for careful reading of our manuscript and for the thoughtful comments and constructive suggestions, which help to improve the quality of this manuscript. Our response follows.

Reviewer #1 Dr Monika Rucinska: Comments to author

Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format.

Please overwrite this text when adding your comments to the authors.
Reply: Because I didn’t find any specific points to be corrected in your comments, our revised was done based on reviewer#2 comments.

Reviewer #2 Dr Alparslan Serarslan: Comments to author 1. In the introduction section, it can be better to give information about heparinoid moisturizer and its mechanism of action.

Reply: Thank you for your comments. I added the information about heparinoid moisturizer in the background section, line 3, page 5.